Author’s response to reviews

Title: Measles Outbreak in an Office Building in the Crowded Metropolis of Beijing, China

Authors:
Zhen Li (foverzhen@126.com)
Zheng Zhang (zhangzheng@cycdpc.org)
Fang Wang (wangfang@cycdpc.org)
Rui Wei (weirui@cycdpc.org)
Jianhong Zhao (zhaojianhong@cycdpc.org)
Fang Liu (liu1210@126.com)

Version: 3 Date: 21 Aug 2019

Author’s response to reviews:

Technical Comments:

Editor Comments:

1. Please confirm whether your study was submitted to and approved by your institutional ethics committee and include a statement to this effect in your “Ethics approval and consent to participate” section. Please also ensure that the full name of your ethics committee is included in this statement. If the need for ethics approval was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this, including the name of the IRB or a reference to the relevant legislation.

Response: We have checked this and added in the manuscript.

2. Please confirm whether informed consent, written or verbal, was obtained from all participants and clearly state this in your "Ethics and Consent to Participate" section. If verbal, please state the reason and whether the ethics committee approved this procedure. If the need for consent was waived by an ethics committee or is deemed unnecessary according to national regulations,
please clearly state this, including the name of the committee or a reference to the relevant legislation.

Response: We have checked this and added in the manuscript.

3. The Author contribution section includes reference to RM, HZ and LL as authors, who are not listed as authors elsewhere. Please clarify whether RM, HZ and LL should be included as an author for this manuscript.

Response: RM and LL were not in the list of author and the name of HZ was misspelling. We have corrected this in the manuscript.

4. The individual contributions of all authors to the manuscript should be specified in the Authors’ Contributions section. Currently the individual contributions of Jianhong Zhao are not included.

Response: The individual contributions of Jianhong Zhao was included in the manuscript and the name was misspelling. We have already corrected the name of Jianghong Zhao.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We uploaded the clean version of manuscript and figure.